Author's response to reviews

Title: Pregabalin versus gabapentin in partial epilepsy: a meta-analysis of dose-response relationships

Authors:

Philippa Delahoy (philippa.delahoy@pfizer.com)
Sally Thompson (sally.thompson@pfizer.com)
Ian C Marschner (ian.marschner@mq.edu.au)

Version: 3 Date: 26 October 2010

Author's response to reviews:

Miss Angelina Ilievksa MSc (on behalf of Dr Ravindra Garg)
Editor
BMC Neurology

Dear Miss Ilievksa,

Please find the proof-revised version of our manuscript. The only changes made were those recommended by your editorial team.

Kind regards

Philippa Delahoy, BSc (Hons)
Pfizer Australia,
38-42 Wharf Road,
West Ryde,
Sydney,
NSW 2114,
Australia
Tel: + 61 2 9850 3333
Fax: + 61 2 9850 3111
E-mail: philippa.delahoy@pfizer.com

E-mail addresses of the other authors:
sally.thompson@pfizer.com